Author’s response to reviews

Title: Attitudes to Fertility and Childbearing Scale (AFCS): An Assessment of a New Instrument for Women Not Yet Mothers in Sweden

Authors:

Malin Soederberg (malin.soderberg@ki.se)
Ingela Lundgren (ingela.lundgren@gu.se)
Kyllike Christensson (kyllike.christensson@ki.se)
Ingegerd Hildingsson (ingeerdr.hildingsson@miun.se)

Version: 3 Date: 21 January 2013

Author's response to reviews: see over
Concerning the manuscript,
Attitudes to Fertility and Childbearing Scale (AFCS): An Assessment of a New Instrument for Women Not Yet Mothers in Sweden

Dear Miss Candice Landrito,

Thanks for your comments, below you will find my answers and in the manuscript the corrections is marked yellow. The main manuscript has been formatted in order with the journal style.

Editorial request:

1. Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

   Answer: In line 153-154 (marked yellow) a sentence has been added to clarify the informed consent

2. Please update your ethics statement to include the name of the ethics committee that approved your study.

   Answer: In line 114-115 you can read the name of the ethics committee, the number of the approved statement has been added (the whole sentence is marked yellow)

Yours sincerely,

Malin Söderberg
PhD student, RNRM